Author’s response to reviews

Title: Septin 9 methylated DNA is a sensitive and specific blood test for colorectal cancer

Authors:

Jorja D Warren (jorja.d.warren@aruplab.com)
Wei Xiong (wei.xiong@aruplab.com)
Ashley M Bunker (ashley.bunker@aruplab.com)
Cecily P Vaughn (vaughnc@aruplab.com)
Larissa V Furtado (larissa.furtado@aruplab.com)
William L Roberts (william.roberts@aruplab.com)
John C Fang (john.fang@hsc.utah.edu)
Wade S Samowitz (wade.samowitz@aruplab.com)
Karen A Heichman (karen.heichman@aruplab.com)

Version: 4 Date: 2 November 2011

Author’s response to reviews:

I have reviewed the document and made a change in the reference citations. It now should read (14-17) instead of (14,15,17). With regard to the consecutive numbering of the tables in the manuscript, the original was fine. I believe that there was a mistaken Table 2 prior to Table 1 due to the fact that when Supplementary Table 2 was referenced, and the "Supplementary" was at the end of a line, with the "Table 2" coming on the following line.